 Vc/27-Sep-04
Week40 starting 27Sept  04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 27 Sept 04  ~16.30 hrs 

	
	A1
	MBBL1148
	CR 1 Leak Test
	

	
	A2
	MBBL1122
	PT 6.8 Training 8
	

	
	B1
	SSS602
	ICS 6 Disconnect Magnet WP07
	

	
	B2
	MBBL3213
	PT 6.2 Training 2
	

	
	C1
	MBAR2037
	ICS 2 Connect Magnet (ICS)
	

	
	C2
	MBAR1142
	Thermal cycle
	

	
	D1
	SSS050
	ICS 1 WP04 HV Test Warm
	

	
	D2
	SSS046
	ICS 6 Disconnect Magnet WP07
	

	
	E1
	MBAR2033
	PT 12 Warm Up
	

	
	E2
	MBBL3220
	Prep 3 CDW1 HV
	

	
	F1
	MBAL2094
	Prep 5 Pumping + Cool Down
	

	
	F2
	MBBL2103
	ICS 5 Traveller Last
	


9/27/2004 4:34:46 PM    Temperature data are taken from TBSM18 site
Coordinator this week: John Miles; MTM support on call :  (16-3602).

10 Magnets [ 8 dipoles( 3 with MM)+ 1 SSS + 1 SSSS]   Disconnected after tests last week ( Sat Midnight to Sat midnight)  
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
Priority settings …
· SSS work :   A & B type clarifications
· Special SSS :
· Cluster C Rotary Switch Modifications : expected to be finished on ? 

RESULTS of Tests LAST WEEK 

	Magnets under Test from 19-09-2004 to 25-09-2004 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA2
	MBAR2038 
	[image: image1.png]



	14-09-2004 
	140 h.
	11 h.
	13 h.
	17 h.
	14 h.
	22-09-2004 
	192 hours
	
	

	2
	TBB2
	MBAR1134 
	[image: image2.png]



	13-09-2004 
	39 h.
	45 h.
	52 h.
	13 h.
	7 h.
	20-09-2004 
	168 hours
	
	

	3
	TBC1
	MBAL3185 
	[image: image3.png]



	15-09-2004 
	16 h.
	67 h.
	37 h.
	15 h.
	4 h.
	21-09-2004 
	144 hours
	
	

	4
	TBC2
	MBAR1142 
	[image: image4.png]



	16-09-2004 
	30 h.
	99 h.
	74 h.
	39 h.
	0 h.
	25-09-2004 
	216 hours
	
	

	5
	TBD1
	SSS045 
	[image: image5.png]



	15-09-2004 
	77 h.
	17 h.
	108 h.
	14 h.
	1 h.
	24-09-2004 
	216 hours
	
	

	6
	TBE1
	MBBL3212 
	[image: image6.png]



	16-09-2004 
	28 h.
	44 h.
	25 h.
	25 h.
	6 h.
	22-09-2004 
	144 hours
	
	

	7
	TBE2
	MBBL3214 
	[image: image7.png]



	15-09-2004 
	25 h.
	46 h.
	19 h.
	42 h.
	5 h.
	20-09-2004 
	120 hours
	
	

	8
	TBF1
	MBAL2088 
	[image: image8.png]



	18-09-2004 
	22 h.
	63 h.
	49 h.
	24 h.
	10 h.
	25-09-2004 
	168 hours
	
	

	9
	TBF2
	MBAL3209 
	[image: image9.png]



	14-09-2004 
	20 h.
	33 h.
	46 h.
	12 h.
	1 h.
	19-09-2004 
	120 hours
	
	


+1 SSSS ? = sss602
Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR2038
	10581 A.
(D1_L)
	12274 A.
(D2-U)
	[image: image10.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAR1134
	11555 A.
(D1 U)
	11714 A.
(D2-L)
	11733 A.
(D2_U)
	11864 A.
(D1-L)
	11980 A.
(D2-U)
	12069 A.
(D1_U)
	12041 A.
(D1-L)
	12124 A.
(D2-U)
	12180 A.
(D1_L)
	[image: image11.png]



	-
	-
	-

	3
	MBAL3185
	11124 A.
(D1-U-S13)
	12521 A.
(D1-U-S13)
	[image: image12.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBAR1142
	11585 A.
(D1-L-S01)
	11909 A.
(D2-U-S13)
	12146 A.
(D2-U-S13)
	12395 A.
(D1-U-S01)
	12573 A.
(D1-U-S01)
	12639 A.
(D2-L-S13)
	12728 A.
(D2_U_S13)
	12706 A.
(D1-L-S13)
	12793 A.
(D1_U_S01)
	[image: image13.png]



	-
	-
	-

	5
	SSS045
	12148 A.
(Q_I_12)
	12850 A.
[image: image14.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBL3212
	11268 A.
(D2-U)
	12857 A.
[image: image15.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBBL3214
	12431 A.
(D2 U)
	12850 A.
[image: image16.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBAL2088
	11515 A.
(D1-L-S13)
	12543 A.
(D1-U-S11)
	[image: image17.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBAL3209
	11075 A.
(D1-U)
	12345 A.
(D2_U)
	[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


Bench status  and tests for now (27/09/2004 16:33)  & next few days :  see Video Display

 

	Magnets under Test on 27-09 4:40:24 PM 

	-
	Bench
	Magnet
	Shafts
	Start
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	Last Test
Desc
	
Time
	
I
	
Temp.
	Total

	1
	TBA1
	MBBL1148 
	[image: image19.png]



	26-09 
	19 h.
	9 h.[image: image20.png]



	-
	-
	-
	CR 1 Leak Test
	27-09
(15 : 21) 
	0 A. 
	300 K
	28 hours

	2
	TBA2
	MBBL1122 
	[image: image21.png]



	22-09 
	20 h.
	54 h.
	43 h.[image: image22.png]



	-
	-
	PT 6.8 Training 8
	27-09
(13 : 21) 
	12574 A. 
	1.9 K
	117 hours

	3
	TBB1
	SSS602 
	[image: image23.png]



	15-09 
	40 h.
	77 h.
	75 h.
	17 h.
	13 h.
	ICS 6 Disconnect Magnet WP07
	24-09
(22 : 48) 
	0 A. 
	300 K
	288 hours

	4
	TBB2
	MBBL3213 
	[image: image24.png]



	20-09 
	42 h.
	117 h.
	17 h.[image: image25.png]



	-
	-
	PT 6.2 Training 2
	27-09
(12 : 10) 
	12375 A. 
	1.9 K
	176 hours

	5
	TBC1
	MBAR2037 
	[image: image26.png]



	27-09 
	2 h.[image: image27.png]



	-
	-
	-
	-
	ICS 2 Connect Magnet (ICS)
	27-09
(14 : 49) 
	0 A. 
	300 K
	2 hours

	6
	TBC2
	MBAR1142 
	[image: image28.png]



	26-09 
	31 h.[image: image29.png]



	-
	-
	-
	-
	Thermal cycle
	26-09
(09 : 11) 
	0 A. 
	300 K
	31 hours

	7
	TBD1
	SSS050 
	[image: image30.png]



	24-09 
	70 h.[image: image31.png]



	-
	-
	-
	-
	ICS 1 WP04 HV Test Warm
	24-09
(23 : 30) 
	0 A. 
	300 K
	70 hours

	8
	TBD2
	SSS046 
	[image: image32.png]



	07-09 
	249 h.
	97 h.
	92 h.
	15 h.
	14 h.
	ICS 6 Disconnect Magnet WP07
	27-09
(08 : 49) 
	0 A. 
	300 K
	475 hours

	9
	TBE1
	MBAR2033 
	[image: image33.png]



	22-09 
	25 h.
	71 h.
	33 h.
	4 h.[image: image34.png]



	-
	PT 12 Warm Up
	27-09
(12 : 11) 
	2 A. 
	Warming
	133 hours

	10
	TBE2
	MBBL3220 
	[image: image35.png]



	27-09 
	8 h.[image: image36.png]



	-
	-
	-
	-
	Prep 3 CDW1 HV
	27-09
(15 : 28) 
	0 A. 
	300 K
	8 hours

	11
	TBF1
	MBAL2094 
	[image: image37.png]



	25-09 
	28 h.
	24 h.[image: image38.png]



	-
	-
	-
	Prep 5 Pumping + Cool Down
	26-09
(16 : 50) 
	0 A. 
	Cooling
	52 hours

	12
	TBF2
	MBBL2103 
	[image: image39.png]



	19-09 
	43 h.
	85 h.
	34 h.
	22 h.
	0 h.
	ICS 5 Traveller Last
	27-09
(14 : 49) 
	0 A. 
	300 K
	186 hours


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL1148
	[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL1122
	11550 Amp.
(D1-L-S01)
	11802 Amp.
(D1-L-S13)
	12248 Amp.
(D1-L-S11)
	12381 Amp.
(D2-L-S13)
	12625 Amp.
(D1-U-S01)
	12644 Amp.
(D2-L-S09)
	12574 Amp.
(D2-L-S09)
	[image: image41.jpg]



	-
	-
	-
	-
	-

	3
	SSS602
	5019 Amp.
(M_I_34)
	5477 Amp.
(Q_E_34)
	5812 Amp.
(Q_I_12)
	5820 Amp.
([image: image42.jpg]


)
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBL3213
	10602 Amp.
(D2_U)
	12375 Amp.
(D1_U)
	[image: image43.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBAR2037
	[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBAR1142
	[image: image45.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS050
	[image: image46.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS046
	12162 Amp.
(QI_34)
	12782 Amp.
(QE-34)
	[image: image47.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBAR2033
	10654 Amp.
(D1-U)
	11633 Amp.
(D1-U)
	12358 Amp.
(D1-U)
	12847 Amp.
(D1-L)
	12850 Amp.
([image: image48.jpg]


)
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBL3220
	[image: image49.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAL2094
	[image: image50.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBL2103
	11254.6 Amp.
(D1-U)
	12413 Amp.
(D2 L)
	[image: image51.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


General Problems last week :  
	Date
	Bench
	Problem
	Type
	Application/soft
	Action/advise

	Tuesday _21_A
	TBC2
	MM: Two coils bad in the shaft:
	MTM
	
	Stephane informed by Federico: he  advised to proceed and report the errors..

	Tuesday _21_A
	TBD1
	Shaft in aperture 2 touched the anticrysotat in Qloc position.

 The cable connecting the shaft to the TRU was too short. No measurement possible in aperture 1.
	MTM
	
	OK to proceed : OK for MM not for Qloc.

Fabienne informed. She changed the cable.

	Wednesday _22_M
	TBD1
	SSS045: IAP Tcoil: Faulty results for sectors 1, 2, 3 in aperture 1: faulty values in aperture 2 also
	MTM
	
	

	Wednesday _22_A
	TBC2
	MM1: 

-Problem in shaft position.

-Current not displayed

-Temperature not correct (7.5 K). .
	MTM
	
	Wrong coil sets

Not displayed in manual. Displayed in automatic?



	Wednesday _22_A
	TBF1
	IAP at cold: compensation out of limits.
	MTM
	IAP
	Federico called:He checked all the chain: nothing wrorng. Proceed as is.



	Wednesday _22_A
	TBD1
	LHC cycle could not be exported as the magnet name was not recognized. 
	MTM
	
	Stephane gave the correct name: corrected by Maryline.

	Thursday_23_M
	TBD1
	MinNRJ done to warm up the magnet but only one training quench performed.
	MTM
	
	TC asked to colldown again the magnet and do a training to warm up the magnet.

	Thursday_23_M
	TBF1
	B11 wrong in one sector (over-range).
	MTM
	
	Stephane advised to proceed..

	Thursday_23_A
	TBF1
	SPA during LHC cycle at 5700 A during the measurement cycle 
	IT
	
	Federico called. Stephane informed. He decided not to repeat the measurement.

	Thursday_23_N
	TBF1
	Cryo problem due to LHe level. 
	ACR
	
	

	Thursday_23_N
	TBD2
	It is not possible to power CD circuit corrector magnet.

Current ramp down after powering EF circuit. 
	MTM
	
	Federico called. Files were missing .He also found a permanent trigger in Vclamp_EF. He put a bridge on the ‘sortie double’of the potaim card.

	Friday_24_N
	TBD1
	HV test failed on SSS050(WP04). bad insulation sextupole/MQT.
	MTM
	
	Federico called : Investigation to see if it come from the IFS box monday.

	Saturday_25_M
	TBC1

TBA2
	Wrong magnet name and shaft name s in MMP.
	MTM
	
	Federico called : He corrected the setting.

	Saturday_25_A

Sunday_26_M
	TBC1


	Over-range in sector 1415 Ap2

Motor position had a problem.
	MTM
	
	Federico called (shaft:30) He checked but the coli is OK. Origin not found. Stephane advised to continue.

Adjusted.

	Saturday_25_M
	TBE2
	2095 magnet: HV report not generated correctly
	MTM
	HV
	Richard informed by E-mail

	Sunday_26_N
	TBB2
	Quench occurred during diode test at 466 A while ramp at 3 kA.

Setting “special magnet” instead of ‘dipole magnet’ used in power application.


	MTM
	
	Power supply application closed , select the good setting and then OK.








